Author's response to reviews

Title: Efficacy and tolerance of the topical application of potassium hydroxide (10% and 15%) in the treatment of molluscum contagiosum. Randomized clinical trial. Research Protocol.

Authors:

Josep R Marsal (jmarsal lleida ics@genca.t cat)
Ines Cruz (icruz lleida ics@genca.t cat)
Concepció Teixidó (cteixido lleida ics@genca.t cat)
Olga Diez (odiez lleida ics@genca.t cat)
Mireia Martinez (mimartinez@arnau scs.es)
Gisela Galindo (ggalindo lleida ics@genca.t cat)
Jordi Real (jreal lleida ics@genca.t cat)
Joan A Schoenenberger (jas@arnau scs.es)
Helena Pera (hpera@idiapigol org)

Version: 2 Date: 9 May 2011

Author's response to reviews: see over
Dear Editor,

Thank you for your response. We have reviewed your comments and made changes in the manuscript, as follows:

1. The manuscript now specifies the approval received by the Research Ethics Committee of the Instituto de Investigación en Atención Primaria IDIAP J Gol (Primary Care Research Institute) and their certification has been sent to you by email. In addition, the document specifying the terms of the grant from the Spanish Ministerio de Ciencia e Innovación (Science and Innovation Ministry), which includes the total monetary support received, is also attached.
2. At the moment the study is in the recruitment phase. We have not submitted this manuscript to any other journal.
3. The study was registered at ClinicalTrials.gov, effective May 4, 2011.
4. A native Anglophone medical translator has reviewed the entire manuscript.
5. Discussion has been added to the Abstract and manuscript.

Thank you for your attention and your feedback. Please do not hesitate to contact us to resolve any doubts or to make any further comments.

Sincerely,

Inés Cruz Esteve
Lleida, 05/05/2011